[[reference-search_application-search]]

////////
===========================================================================================================================
||                                                                                                                       ||
||                                                                                                                       ||
||                                                                                                                       ||
||        ██████╗ ███████╗ █████╗ ██████╗ ███╗   ███╗███████╗                                                            ||
||        ██╔══██╗██╔════╝██╔══██╗██╔══██╗████╗ ████║██╔════╝                                                            ||
||        ██████╔╝█████╗  ███████║██║  ██║██╔████╔██║█████╗                                                              ||
||        ██╔══██╗██╔══╝  ██╔══██║██║  ██║██║╚██╔╝██║██╔══╝                                                              ||
||        ██║  ██║███████╗██║  ██║██████╔╝██║ ╚═╝ ██║███████╗                                                            ||
||        ╚═╝  ╚═╝╚══════╝╚═╝  ╚═╝╚═════╝ ╚═╝     ╚═╝╚══════╝                                                            ||
||                                                                                                                       ||
||                                                                                                                       ||
||    This file is autogenerated, DO NOT send pull requests that changes this file directly.                             ||
||    You should update the script that does the generation, which can be found in:                                      ||
||    https://github.com/elastic/elastic-client-generator-js                                                             ||
||                                                                                                                       ||
||    You can run the script with the following command:                                                                 ||
||       npm run elasticsearch -- --version <version>                                                                    ||
||                                                                                                                       ||
||                                                                                                                       ||
||                                                                                                                       ||
===========================================================================================================================
////////

[discrete]
[[client.searchApplication.search]]
=== client.searchApplication.search

Run a search application search. Generate and run an Elasticsearch query that uses the specified query parameteter and the search template associated with the search application or default template. Unspecified template parameters are assigned their default values if applicable.

{ref}/search-application-search.html[{es} documentation]

[discrete]
==== Function signature

[source,ts]
----
(SearchApplicationSearchRequest, options?): Promise<SearchApplicationSearchResponse>
----

[discrete]
==== Request

[pass]
++++
<pre>
++++
interface SearchApplicationSearchRequest extends <<RequestBase>> {
  name: <<Name>>
  typed_keys?: boolean
  params?: Record<string, any>
}

[pass]
++++
</pre>
++++
[discrete]
==== Response

[pass]
++++
<pre>
++++
type SearchApplicationSearchResponse<TDocument = unknown, TAggregations = Record<<<AggregateName>>, <<AggregationsAggregate>>>> = <<SearchResponseBody>><TDocument, TAggregations>

[pass]
++++
</pre>
++++
